Early-Life Economic Conditions and Old-Age Mortality: Evidence from Historical County-Level Bank Deposit Data^{*}

Valentina Duque[†]

Jason Fletcher[‡]

Vikas Gawai[§]

Hamid Noghanibehambari**

Lauren Schmitz^{††}

Abstract

This paper studies the long-run mortality effects of in-utero and early-life economic conditions. We examine how local economic conditions experienced in the Great Depression, proxied by county level banking deposits during in-utero and first years of life, can influence old-age longevity. We find that a one-standard-deviation rise in per capita bank deposits is associated with an approximately 1.7 months increase in longevity at old age. The effects are robust across a wide array of specification checks. Additional analyses comparing state-level versus county-level economic measures provide insight on the importance of controlling for local-level confounders and exploiting more granular

^{*} The authors claim that they have no conflict of interest to report. The authors would like to acknowledge financial support from NIA grants (R01AG060109, R01AG076830, K99AG056599 R00AG056599), the Center for Retirement Research Steven H. Sandell Grant Program pursuant to a grant from the U.S. Social Security Administration (BC20-S2), and the Center for Demography of Health and Aging (CDHA) at the University of Wisconsin-Madison under NIA core grant P30 AG17266. The content is solely the responsibility of the authors and does not necessarily represent the official views of the Social Security Administration or the NIA. We thank members of the UW Health Economics Working Group for helpful comments.

[†] Department of Economics, University of Sydney, FASS Building A02, Room 564 University of Sydney, NSW, 2006, Australia

Email: valentina.duque@sydney.edu.au

[‡] La Follette School of Public Affairs, University of Wisconsin-Madison, 1225 Observatory Drive, Madison, WI 53706-1211, USA

Email: jason.fletcher@wisc.edu

[§] Department of Agricultural & Applied Economics, University of Wisconsin-Madison, 305 Taylor Hall, 427 Larch St., Madison, WI 53706, USA

Email: gawai@wisc.edu

^{**} Center for Demography of Health and Aging, University of Wisconsin-Madison, 1180 Observatory Drive, Madison, WI 53706

Email: noghanibeham@wisc.edu, Phone: +1-806-620-1812, ORCID ID: https://orcid.org/0000-0001-7868-2900

^{††} La Follette School of Public Affairs, University of Wisconsin-Madison, 1225 Observatory Drive, Madison, WI 53706-1211, USA

Email: <u>llschmitz@wisc.edu</u>

measures in exploring the relationship between early-life conditions and later-life mortality.

Keywords: Mortality, Longevity, Great Depression, Historical Data **JEL Codes**: H75, I12, I18, J18

1. Introduction

Increases in life expectancy have been one of the most significant improvements in households' welfare over the last century (Cutler et al., 2006; Deaton, 2003).⁷ In the past 100 years, Americans have enjoyed an overall gain of approximately 25 years in life expectancy at birth. Motivated by the growing body of research showing that conditions experienced in utero and early childhood are key determinants of health and human capital outcomes later in life (Almond et al., 2018; Almond & Currie, 2011; Barker, 1990), we provide new evidence on the relationship between early-life economic conditions and old age mortality, by focusing on the most severe economic recession in American history, the Great Depression.

Using micro-level data on the universe of deaths from 1975 and 2005 obtained from the Social Security Administration death records and linked with the 1940 U.S. Complete Count Census, this paper analyzes how individual longevity is affected by local economic shocks before birth and during the first years of life. We proxy local economic conditions with annual, countylevel per capita bank deposits.⁸ We focus on this measure because financial distress in the banking system played a major role in propagating the contraction of economic activity and employment during the Great Depression (Bernanke, 1983; Fisher, 1933; Friedman & Schwartz, 1963). Therefore, our measure captures the severity of the economic shock experienced by households

⁷ Several explanations have been provided in the literature for such a staggering improvement, including rising incomes and economic development (Acemoglu & Johnson, 2007; Costa, 2015; Fogel, 1994; Preston, 1975), the introduction of new drugs and scientific innovations (Bleakley, 2007), and/or public health investments (Anderson et al., 2022; Cutler & Miller, 2005).

⁸ The deposits reflect both the supply and the demand of credit in the local area. For instance, an economic downturn destroys local jobs and reduces earnings and subsequent savings. On the other hand, it also affects market expectations about the future of the economy and influences the decisions of firms in their demand for credit. However, we show that the equilibrium quantity co-moves strongly and significantly with alternative measures of the economy such as income and retail sales

across different regions while, unlike all other measures of economic conditions used in the literature, having the advantage of being available at the county and year level prior to 1929.

Our empirical model uses a two-way fixed effects strategy that compares the longevity of cohorts born before, during, and after the Great Depression and who were differentially exposed to local economic shocks around the time of birth, controlling for county-level time-invariant characteristics, temporal shocks, and within-state county-group-by-year fixed effects. Our findings suggest that a one-standard-deviation increase in bank deposits during the in-utero period, which is equivalent to roughly 4 times the drop in deposits between the years 1929-1933 (peak-to-trough of the Great Depression), is correlated with a roughly 1.7 months higher age at death in old age. The effect is quite robust across a wide array of specifications. To assess whether these relationships are an artifact of overall trends in health improvements/disruptions, we implement a placebo test and show that the effects become indistinguishable from zero for deposits at preprenatal ages. We also argue that these effects are not driven by endogenous demographic changes in the sample due to changes in fertility, early-life survival, or migration. Additional analysis suggests that improvements in educational attainments and income in adulthood are potential mechanism channels. However, using estimates from previous studies, we posit that these channels only partially reflect the pathways.

The motivation of the current study stems from the limited evidence of in-utero economic conditions and later-life longevity. Van Den Berg et al. (2006) were the first to document the adverse effects of national economic conditions around birth on life expectancy using historical data for the Netherlands. The authors found that cohorts born during an economic boom lived 1.6 years longer (or 4 percent longer relative to the life expectancy of 39 years) than those born during economic recessions. No effects were found when booms were experienced during early

childhood. Similarly, Lindeboom et al. (2010), also using data from the Netherlands, showed that children born during the Potato famine of the mid-nineteen hundreds lived 2.5-4 fewer years as adults compared to those born before the nutritional shock and with more pronounced impacts for children from lower class families.

We contribute to the literature in several ways. First, we contribute to a small literature that analyzes the link between economic conditions and mortality in the context of the Great Depression that has found mixed results. While Granados et al (2009) showed a negative correlation between the GDP per capita and the national mortality rate, Stuckler et al. (2012) found no effect between changes in bank suspensions and changes in mortality except for an increase in suicide rates. Fishback et al. (2007) in contrast, showed a small decline in death rates during the 1930s due to increases in New Deal spending. Cutler et al. (2007) found little evidence that early life exposure to the Depression affected long-term health (including mortality) using longitudinal data from the Health and Retirement Study (HRS) linked to regional-level macroeconomic data. More recently, however, Schmitz & Duque (2022) and Duque & Schmitz (2021) revisited this question in the HRS using macroeconomic data linked to the state of birth and found improvements in the magnitude and precision of the effects on old age health and mortality when economic outcomes were measured at the state level as opposed to the regional level. Importantly, these effects were localized to the in utero period specifically as opposed to the pre-conception, postnatal, childhood, or early adolescent periods. Thus, from an empirical perspective, we also contribute to the literature by using longitudinal bank deposit data measured at the county level before and during the Great Depression to explore within-state geographic variation in economic conditions, as opposed to prior studies that relied on region-level and state-level variations in the shock. In addition, our data source contains millions of observations, which adds power to our statistical tests and allows the research design to search for potential heterogeneity in the effects across different demographic groups.⁹

The rest of the paper is organized as follows. Section 2 reviews the literature. Section 3 introduces data sources. Section 4 discusses the empirical framework. Section 5 reviews the results. Section 6 explores endogenous fertility. Section 7 suggests potential mechanism channels. Section 8 concludes the paper.

2. Literature Review

A growing body of research evaluates the long-term effects of early life adversities (Almond et al., 2018; Almond & Currie, 2011; Currie, 2009; Hayward & Gorman, 2004; Steptoe & Zaninotto, 2020). Several studies provide suggestive evidence for the relevance of in-utero and early-life conditions for short-term and later-life outcomes, including child mortality (Baird et al., 2011; Schoeps et al., 2018), cognitive development (Chang et al., 2022; Majid, 2015; Yamashita & Trinh, 2022), test scores (Almond et al., 2015; Shah & Steinberg, 2017), education (Aizer, Stroud, et al., 2016; Caruso & Miller, 2015), adulthood earnings (Black et al., 2007; Currie & Rossin-Slater, 2015; Hoynes et al., 2016), and health outcomes over the life cycle (Fletcher, 2018; Fletcher et al., 2010; Fletcher & Noghanibehambari, 2022; Goodman-Bacon, 2021; Miller & Wherry, 2019; Noghanibehambari & Engelman, 2022; Persson & Rossin-Slater, 2018). Early-life conditions could operate through these mediatory pathways to influence the trajectory of old-age longevity. For instance, den Berg et al. (2015) employ a longitudinal panel of observations in Dutch registries covering about two centuries and showed that men who were born during an

⁹ Our paper also contributes to an emerging literature that has looked at the role of economic resources–through specific welfare programs—on longevity. Two recent studies by Aizer, Eli, et al. (2016) and Aizer et al. (2020) that focused on the introduction of the U.S. Mother's Pension program in 1937 and the largest youth training program in history implemented during the New Deal, respectively, showed significant effects of these government interventions on beneficiaries' life expectancy.

economic boom (versus a recession) are more likely to be married during adulthood and at old ages and have a lower risk of mortality. They argue that, among men, marriage has a protective effect against mortality. Grimard et al. (2010) use data from Mexico and show that socioeconomic status measures during childhood significantly affect old-age health outcomes even after accounting for education and income. Bengtsson & Broström (2009) use data from Sweden and show that early life disease loads affect old-age mortality and socioeconomic status. However, they do not find evidence that the early-life health environment effect on later-life mortality operates through wealth and socioeconomic channels. Gagnon & Bohnert (2012) employ data from Canada and show that family wealth and the socioeconomic status during early life affect mortality during old ages among males. Their results fail to provide evidence of this association among females.

Hayward & Gorman (2004) use data from the National Longitudinal Survey of Older Men and show that men's mortality is correlated with an array of early-life and childhood conditions, including parents' socioeconomic status, mother's marital status, mother's labor force status, and parents' nativity. Montez & Hayward (2011) use the Health and Retirement Study to explore earlylife family socioeconomic status on later-life mortality. They find significant positive correlations between risks of mortality during adulthood and a series of early-life adversities, including perceived poverty during childhood, having a low-educated father, and self-reported poor childhood health. On the other hand, Myrskylä (2010) finds weak and modest effects of early-life conditions on adult mortality using data from several developed European countries. The results suggest a strong correlation between period effects and mortality rather than early-life effects.

Studies that examine the role of economic conditions on later-life mortality outcomes exploit measures of economic conditions at various levels of aggregation and find different results.

For instance, Van Den Berg et al. (2006) use historical longitudinal data from the Netherlands to explore economic conditions in early life on old-age mortality. They exploit the cyclicality of national Gross Domestic Product (GDP) as a proxy for economic conditions and find that being born during a boom versus a recession results in 8 percent lower mortality rates. Arthi (2018) explores the persistent effects of in-utero and childhood exposure to state-level measures of the Dust Bowl, a devastating environmental shock followed by agriculture failure and reductions in income, on later-life human capital and health. The result suggests long-lasting effects on income, disability, and college completion. Similarly, Duque & Schmitz (2021) and Schmitz & Duque (2022) find a connection between state-level in utero exposure to wages, employment, and car sales during the Great Depression and late-life aging outcomes and mortality. However, an earlier paper by Cutler et al. (2007) that used exposure to economic variation at the census-region level during the Great Depression failed to find any evidence that fetal exposure to economic conditions was associated with disability and chronic disease later in life. Atherwood (2022) implements county-level Dust Bowl measures and explores the effects of young adulthood exposure on laterlife longevity. He finds insignificant average effects. Noghanibehambari & Fletcher (2022) reexamine the effects of early-life exposure to the Dust Bowl on old-age longevity. They employ difference-in-difference method and account for county-level heterogeneity. They find intent-totreat effects of about 1 month reduction in longevity among cohorts that were severely affected by their county-of-birth topsoil erosion in early life.

3. Data and Sample Selection

The primary data source is the Social Security Administration Death Master File (hereafter DMF) records extracted from the CenSoc Project database (Goldstein et al., 2021). The DMF data contains death records among males between 1975-2005 that are linked to the full-count 1940

census. Therefore, they have a wide array of early-life social and economic variables. There are three advantages of DMF data over similar data sources that contain mortality information necessary for our research design. The availability of county identifiers in the 1940-census allows for more granular and detailed environmental information as opposed to virtually all other data sources with state identifiers. Second, the DMF builds a longitudinal panel that contains millions of observations while similar longitudinal data provides several thousand (e.g., Health and Retirement Study). Third, the DMF-census-linked data offers a wide array of family-level covariates, including parental education and a socioeconomic score that can be used in our balancing tests, in analysis of heterogeneous impacts by family resources, and adds the robustness to our identification strategy.

We proxy local economic conditions using changes in bank deposits compiled by the Federal Deposit Insurance Corporation (FDIC) and taken from Manson et al. (2017). The data reports total annual deposits in all state and federal banks in each county (except Wyoming and DC) over the years 1920-1936 as of December of each year. Our choice of this proxy is based on two facts. First, later in the paper, we will illustrate the positive association between changes in bank deposits and similar proxies of state-level and county-level economic covariates. Second, similar studies show an association between banking crises and city-level economic conditions during a similar period (Stuckler et al., 2012).

To infer county of birth from the DMF-census sample, we take two approaches. First, we use cross-census linking rules provided by the Census Linking Project to merge 1940-census records with 1930-census. In our sample, we are able to match 48 percent of cohorts born 1926-1930 to their 1930 census records. For these matched records, we use county-of-residence in 1930 as county-of-birth. For unmatched cohorts of 1926-1930, and for all cohorts of 1931-1936, we

continue to infer the county-of-birth using the second approach. Specifically, we use information from three variables reported in 1940-census. First, we use information on the county of residence in 1935 as the benchmark proxy for location of birth using the fact that the 1940 Census reports the migration status from five years prior. Second, we use county of residence in 1940 if the individual reported having stayed in the same house since 1935. Third, if the migration status is missing and the person's state of birth is the same as state of residence in 1940, we again use county of residence in 1940 as the proxy for the county of birth. To further reduce the migration issue, we limit the sample to children up to 15 years old as older children usually leave their original household. Doing so will limit the sample to cohorts born after 1926.

Since our purpose is to explore in-utero exposures, we calculate a weighted average of deposits for the nine months before birth, assuming an average of nine months of gestation. In so doing, we assign the current year of deposits to our in-utero deposit measure for months of October through December. For other months, the in-utero measure is a weighted average of current and previous year's deposits, where weights are based on the number of months of a year that overlap with the last nine-month period. For instance, March deposits takes three-ninths of the current year's deposits and six-ninths of the previous year's deposits. We then merge DMF-census sample to bank deposit data based on inferred county-of-birth and the weighted average of previous nine-month deposits.

To control for other county-level sociodemographic changes, we use full-count decennial censuses from Ruggles et al. (2020) for the decennial years 1920-1940. We then linearly interpolate covariates for inter-decennial years. Moreover, for the analysis to explore the association of bank deposits with other economic variables, we use county-level retail sale per capita and state-level income per capita extracted from Fishback et al. (2007).

The final sample includes 1,221,113 individuals from 3,042 counties in 47 states born between 1926-1936 who died between 1975-2005.¹⁰ Summary statistics of the final sample are reported in Table 1. The average per capita bank deposit is \$331. Over the sample period, roughly 10 percent of individuals are born in counties that experienced a 5 to 10 percent drop in total banking deposits relative to the county-specific previous year's value. About 23 percent of individuals live in counties that experienced a drop of more than 10 percent in deposits. The top and middle panels of Figure 1 depict the cross-sectional geographic distribution of per capita deposits and per capita retail sales. There is a visual correlation between our proxy for economic conditions (bank deposits) and a measure of local consumption expenditure (retail sales per capita). The bottom panel of Figure 1 illustrates the distribution of age-at-death by county-of-birth for cohorts born in 1930. There is also a visual correlation between these two measures and long-run longevity. In a cross-sectional and correlational manner, Figure 3 depicts the differences in the density distribution of age at death in the subsample of above-median per capita deposit (in green) versus below median per capita deposit (in red). While these are suggestive figures, they do not convey any informative interpretation of the statistical association.

4. Econometric Method

Our identification strategy exploits within-county and over-time variations in bank deposits. Specifically, we implement regressions of the following form:

$$DA_{icsb} = \alpha_0 + \alpha_1 PCBD_{csb^*} + \alpha_2 X_{icsb} + \alpha_3 Z_{csb} + \xi_c + \zeta_b + \eta_{sb} + \varepsilon_{icsb}, \tag{1}$$

Where the outcome is age at death (DA) of individual *i* born in county-of-birth *c* State Economic Area *s* and birth year *b*. State Economic Areas (SEA) are geographic boundaries that

¹⁰ The bank deposit data does not include data for Wyoming and DC. Also, the 1940 census does not include Alaska and Hawaii. These states are therefore omitted from the sample.

covers several counties within the same state that have similar economic and demographic conditions (Bogue, 1951). SEA was introduced in census 1950 and then was applied to counties in 1940 census. Since the period of the Great Depression was accompanied by vast changes in economic conditions, we prefer exploiting within-SEA across-counties variations to better isolate the impacts of economic conditions. The parameter PCBD represents per capita bank deposits assigned to each individual based on county-of-birth and the average of nine months leading to birth (b^*) . To ease the interpretation, we standardize this variable with respect to the mean and standard deviation of the sample. In X, we include as individual controls dummies for race, gender, and ethnicity. The matrix X also contains parental characteristics, including dummies for maternal education, paternal education, and socioeconomic status. The parameter Z represents a series of county-by-birth year covariates constructed based on full-count decennial censuses 1920-1940 and interpolated for inter-decennial years. These covariates include the share of homeowners, the share of married people, and the average occupational income score. The county fixed effects, represented by ξ , control for time-invariant unobserved features of counties. To account for temporal cohort-level changes in longevity, we add birth cohort fixed effects, represented by ζ . To account for all SEA-by-year divergence in the outcome and other time-varying local determinants, we include SEA-by-birth-year fixed effects represented by η . Therefore, the identifying variation comes from changes in bank deposits across counties within an SEA-year. Finally, ε is a disturbance term. We cluster standard errors at the county level to account for serial correlation in the error term. The coefficient of interest is α_1 that, conditional on covariates and fixed effects, captures the effect of one-standard-deviation (from mean) change in per capita bank deposits on later-life old-age longevity.

5. Results

5.1. Endogeneity Concerns

One potential concern in our analysis is that economically improving areas attract more people and induce migration. Similarly, a recession may affect different areas to varying degrees and generates in/out-migration. If certain characteristics in migrant subpopulations correlate with their later-life health and longevity, the link between deposits and longevity is contaminated by endogeneity. For instance, if whites are more (less) likely to migrate after a county is hit by a recession, the coefficients of equation 1 can overstate (understate) the true effects since whites have higher longevity for reasons not necessarily captured by a race dummy. To explore this source of bias, we ask whether changes in bank deposits are associated with changes in observable characteristics, conditioning on county and SEA-year fixed effects. The results of such balancing tests are reported in Table 2. We do not observe any statistical association between bank deposits and probability of being white, mother's schooling, father's schooling, and father's socioeconomic score. Moreover, the estimated effect sizes are economically small. For instance, based on percent change from the mean of the dependent variable reported in the fifth row, the effects suggest that a one-standard deviation change in bank deposits is correlated with 0.27 percent change in white, 0.15 percent change in maternal schooling, and 0.32 percent change in paternal socioeconomic score. Therefore, the big picture extracted from these numbers is a failure in finding robust and strong evidence of demographic and socioeconomic changes due to changes in bank deposits that could unbalance the sample by contaminating the long-run relationships.

Another concern is that linkage rates between DMF and the 1940-census may be predicted by early-life shocks, such as changes in deposits. While the linking rules are primarily based on name commonality, and we have little prior concern for this being correlated with local economic conditions, we explore merging issues empirically by evaluating the correlation between the merging of DMF-census records and deposits. In so doing, we start by imposing sample selections as discussed in section 3. We also follow the procedure described in section 3 to infer county-ofbirth. The sample selections result in 22,285,131 observations before merging with the DMF death records. The successful merge dummy takes a value of one if DMF is merged with the census records and zero otherwise. We then regress this variable on deposits per capita, conditional on a full specification of equation 1. The results are reported in Table 3 for the full sample, sample of whites, sample of people with low educated mothers, and sample of persons with low socioeconomic score fathers, in columns 1-4, respectively. We find small and insignificant coefficients between deposits and successful merging. For instance, a one-standard-deviation change in banking deposits per capita results in an insignificant 0.8 basis point increase in the probability of merging, equivalently 0.15 percent change from the mean of the outcome.

5.2. Main Results

The main results of the paper are reported in Table 4. We start with a model that only includes county and SEA-by-birth-year fixed effects in column 1 and gradually add additional covariates to the model across consecutive columns. The marginal effect of bank deposits per capita is virtually unchanged across specifications. Since there is enough evidence to believe that race/gender (as individual covariates), mother's education and father's socioeconomic index (as parental characteristics), and county-level demographic and economic covariates are significant determinants of long-run health outcomes, the fact that coefficients do not change across columns lends credibility to the exogeneity of bank deposits, at least with regards to the observable and available variables. The results of the full specification of column 4 suggest that, on average, a one-standard-deviation change (from the mean) in per capita deposits in utero is associated with 1.7 months higher longevity during old ages. We put this number into perspective by comparing it with the coefficients of other variables. For instance, the marginal effect of a black dummy (not

reported here) is -15.3 (se=1.1). Therefore, a one-standard-deviation change in bank deposits is equivalent to roughly 11 percent of the black-white gap in longevity. The difference in average life expectancy between the US and other OECD countries is 49.2 months (76.6 years versus 80.7 years, respectively). Thus, the impact of a one-standard-deviation rise in bank deposits is equivalent to roughly 3.5 percent of the US-OECD-countries gap in longevity.

Another way to understand the magnitude of the finding is to compare with other determinants of longevity and the impact of other early-life exposures. For instance, Chetty et al. (2016) investigate the income-longevity relationship in the US using individual tax returns linked with death records. They document that each 5-income-percentile is associated with about 0.8 months higher age-at-death. Therefore, the effect of a one-standard-deviation rise in bank deposits in county-year-of-birth is equivalent to the impact that moving up the income ladder by about 1 percentile has on longevity. Halpern-Manners et al. (2020) examine the education-mortality relationship using twin fixed effect strategy. They document that each additional year of education is associated with about 4 months higher longevity. Therefore, the marginal effect of Table 4 is equivalent to the effect of 0.43 years of schooling on longevity. Aizer, Eli, et al. (2016) examine the impacts of the Mother's Pension (MP) program, a state-local government joint initiative to help poor single mothers with cash transfers prior to social security era, on children's old-age longevity. They find treatment-on-treated effects of about 1-year additional life to children whose mothers were selected for the MP benefits. The MP benefits usually lasted for three years and transferred about 30-40 percent of pre-transfer maternal income. Therefore, our finding is equivalent to a onetime transfer of 13-17 percent of income to poor families.

5.3. Effects during Pre-prenatal and Postnatal Periods

While the literature points to the relevance of conditions in utero for later-life outcomes, several studies also show associations between post-birth exposures and conditions and long-run outcomes (Chyn, 2018; Currie & Rossin-Slater, 2015; Ludwig & Miller, 2007). This section complements the main results by investigating the effects of bank deposits experienced during postnatal ages and later-life longevity. Moreover, we also examine the association between changes in deposits during pre-prenatal period on longevity. The idea is that if the deposits are capturing general improvements in health conditions rather than in-utero economic condition shocks, we would observe similar effects for the exposure measures assigned for pre-prenatal period. Therefore, this test provides a placebo check to assess the validity of the main results.

In so doing, we generate two new variables for assigning deposits at two years up to gestation and two years after birth. We include these two variables as well as our primary in-utero exposure measure in the full specifications of equation 1 and allow for the effects of these variables to compete within a regression. The results are reported in Table 5. Across specifications, the coefficients of the pre-prenatal development period suggest small and insignificant correlations providing support for the empirical strategy. The correlations between bank deposits and postnatal ages are also small in magnitude (relative to the deposit's assignment during in-utero), and their coefficients are statistically insignificant. The effect of exposure during in-utero suggests positive, relatively large, and statistically significant association between deposits and old-age longevity.

5.4. Exploring the Relevance of County-Level Variations

As we discussed in section 1, an advantage of our study compared to the previous literature is in part that we use a more granular measure of local economic conditions (i.e., county-level) versus national, census-region, or state-level measures of other studies (Arthi, 2018a; Atherwood, 2022; Cutler et al., 2007; Van den Berg et al., 2015; Duque & Schmitz, 2021; Granados et al., 2009; Myrskylä, 2010b; Schmitz & Duque, 2022; Scholte et al., 2015; Van Den Berg et al., 2006; Van den Berg et al., 2009). To show that this granularity is essential in this context, we explore the correlations of economic conditions at the state level with longevity in our sample. Specifically, we use state-level income per capita and deposits per capita aggregated at the state level. Since the state-level income per capita is available only from 1929, we restrict the sample to cohorts of 1929-1936. In all regressions of this section, we include individual and family covariates.

As reported in column 1 of Table 6 the unconditional correlation between state-level income per capita and death age is 9.4 months. However, a large portion of the observed correlation can be explained by unobserved state and cohort characteristics. Controlling for state and cohort time-invariant confounders reduces the coefficient by about 86 percent (column 2). Next, we aggregate the deposits from the county level to the state level and show the correlations with age at death in columns 3-4.¹¹ The correlation is 35.5 months (column 3). The marginal effect becomes smaller (about the same 80% reduction as in Columns 1 and 2) and noisy once we include state and cohort fixed effects (column 4).

In the next step, we use county-level deposits per capita and replicate the results across different specifications. In column 5, we show that the correlation (excluding fixed effects) is about 0.4 months and statistically insignificant. Adding state-fixed effects provides a negative and small coefficient. These results suggest the existence of state and county level variables correlated with bank deposits that have offsetting effects, again suggesting the need to control for county level effects and focus on within-county changes in economic conditions (as measured by bank deposits).

¹¹ We use county-level per-capita deposits and use a weighted average of this value for each state where weights are mean county population over the sample period.

In column 7, we include county fixed effects. The correlation between county level economic circumstances at the time of birth and later life longevity becomes positive and relatively large in magnitude. In column 8, we control for other local-level confounders by including SEA-year fixed effects. The results suggest an increase of 2.9 months due to a one-standard-deviation change in bank deposits.

5.5. Validity of Bank Deposits as a Proxy for Economic Conditions

To gauge the magnitude of the paper's main results and to explore the validity of bank deposits as a proxy for local economic conditions, we explore the relationship between per capita deposits and other local and state-level economic variables. The main limitation of this exercise is the scarcity of local-level data for the time period of this study. We know of no dataset that contains measures of county-level income or county-level unemployment rates during this time period (and that starts prior to the Great Depression). However, income data is available at the state level for the years 1929-1940. Also, retail sale data is available at the county level which is, arguably, a reasonable measure of consumption. These data are available for post-1929 years and are taken from Fishback et al. (2007). Thus, the analysis sample for both retail sales and income cover the years 1929-1936 since our sample ends in the year 1936.

We merge per capita retail sales at the county and year level with our bank deposit sample and implement regressions similar to equation 1. The results are reported in columns 1-3 of Table 7. We start by showing the unconditional correlation in column 1, adding county and year fixed effects in column 2, and then implementing a full specification in column 3. The unconditional correlation suggests a very strong co-movements between bank deposits and retail sales. However, fixed effects explain a large portion of this correlation. The correlation of the full model of column 3 that includes SEA-by-year fixed effects is still statistically and economically significant. A onestandard-deviation rise in per capita deposits is associated with 0.15 standard-deviations increase in per capita retail sale.

For the state-level income analysis, we aggregate our final sample at the state-level and merge it with income data at the state-year level and implement regressions that include state, year, and region-by-year fixed effects. These results are reported in columns 4-6 of Table 7. The full specification of column 6 implies that for a one-standard-deviation change in deposits per capita the income per capita changes by 0.4 standard deviations. Between the years 1929 and 1933 (peak to trough of the Great Depression), income per capita decreased from \$611 to \$326, a decrease of about 1.5 times its standard deviation over the sample period. Using figures from Table 7 and this drop in income, one can deduce a roughly 6.1 months drop in later-life longevity.¹² This is an economically large effect.

5.6. Robustness Checks

In Table 8, we explore the robustness of the main results to alternative specifications. In column 1, we replicate the results of the full specification of Table 4 as the benchmark comparison. In column 2, we allow for the time-invariant effects of counties to vary by individual covariates. In column 3, we add county-by-parental-characteristics fixed effects so that the unobserved time-invariant features of a county can be absorbed differently by families with different sociodemographic backgrounds. The resulting marginal effects are almost identical to the main results. In column 4, we control for all unobserved county characteristics that evolve linearly over cohorts by including county-by-birth-year linear trend. The effect rises and remain statistically significant.

¹² This number is calculated using the marginal effect of column 4 of Table 4 (1.7), the marginal effect of column 6 of Table 7 (0.42), and the change of 1.5 standard-deviations in state-level income, as follows: $\left(\frac{1}{0.42}\right) \times 1.5 \times 1.7$.

In column 5, we control for seasonality in birth and death outcomes by adding birth-month and death-month fixed effects. The resulting coefficient is comparable to that of column 1.

In column 6, we check for robustness of the functional form by replacing the outcome with the log of age at death. The effect suggests 0.2 percent change in the outcome as a result of one-standard-deviation change in deposits, an effect that is almost identical to the percent change effect shown in column 4 of Table 4. In column 7-8, we replace the outcome with a dummy to indicate longevity beyond age 70 and 65, respectively. A one-standard-deviation increase in in-utero bank deposits per capita is associated with 62 and 53 basis-points higher likelihood of living beyond age 70 and 65. These effects are equivalent to roughly 2.6 and 1.1 percent change from the mean of their respective outcomes.

In columns 9, we check for sensitivity of county-level clustering. We find that the errors are larger when we implement a two-way clustering by county and state-year. However, the resulting coefficient is still statistically significant at the 5 percent level.

5.7. Alternative Measures

In Table 9, we replicate the main results using alternative measures of banking conditions. In column 1, we ignore the differences in county population as a deflating channel for the effects of deposits and use total deposits as the main independent variable. The resulting marginal effect suggests 2.2 months higher longevity as a result of a one-standard-deviation change in deposits. In column 2, we replace the independent variable with two dummy indicating that banking deposits in a specific county and year have dropped between 5-10 and more-than-10 percent relative to the county's previous year's deposits, candidate measures of the banking crisis. The result suggests a reduction of 0.6 months in longevity for both measures of banking crisis. We should note that 33 percent of cohorts have experienced a banking crisis during their prenatal period, per our definition of crisis (see Table 1). Overall, these results add to the general picture that the economic conditions of early life have significant effects on later-life longevity.

5.8. Heterogeneity by Subsamples

In Appendix A, we replicate the main results across two subpopulations: whites and blacks. We find that the longevity of black people is more strongly connected with economic conditions. The marginal effect of black sample is roughly 2.5 times that of the white sample, although the effects in both samples are significant.

During the period of the study and specifically for post-1934 years, many counties in the Southern Plains region experienced the Dust Bowl. We examine how the effects vary by Dust Bowl exposure counties in Appendix C. We find that for counties exposed to the Dust Bowl, the correlations are about five times larger, though statistically insignificant. The effect on other counties is almost identical to that of the main results of the paper.

5.9. Sensitivity to Death Window and Gender Selection

The DMF reports death records for males in the years from 1975 to 2005. As an alternative source of data that covers both genders, we use Numident death records of the Social Security Administration extracted from the Censoc Project (Goldstein et al., 2021). Numident is also linked to the 1940 census but reports the death to both females and males for death years 1988-2005. We explore whether the results are sensitive to gender selection and death selection of DMF in Appendix B. We show that when we restrict the sample to Numident death years (i.e., 1988-2005) the effect drops by about 65 percent (column 2 of Appendix Table B-1). This is quite comparable to the Numident results (column 5 of Appendix Table B-1). Therefore, the effects are larger as we expand the death window to cover earlier deaths, suggesting that the Great Depression accelerated the age of mortality.

Looking at the male subsample of Numident reveals an effect of 0.8 months while the female subsample suggests an insignificant effect of 0.2. Therefore, the results are primarily driven by males, suggesting that early-life economic conditions are more relevant to the health of males than females. This fact is in line with studies that show the exposures in early-life are more impactful for males (Clark et al., 2021; Clay et al., 2019; Rosa et al., 2019; Smith et al., 2011; Wang et al., 2017; Weinberg et al., 2008).

6. Fertility Response

While the balancing tests of Table 2 are inconsistent with strong demographic compositional changes due to the differences in the survival of subpopulations, one may be concerned that parents observe the economic condition and plan their fertility accordingly (Currie & Schwandt, 2014; Schaller, 2016; Schaller et al., 2020). However, the literature on economic conditions and fertility is not conclusive (Black et al., 2013; Cohen et al., 2013; Docquier, 2004). Moreover, little empirical research has been done for our study period (Fishback et al., 2007). To explore the selective fertility of parents to the observed deposit changes, we use county-level births data over the years 1926-1936 extracted from Bailey et al. (2016). The data offers three main variables that are specifically useful for the analysis of this section: general birth rate, share of births to whites, and share of births to blacks. Since over time more counties appear in the sample and the effects may be driven by differential fertility of new counties, we balance the county-year sample so that each county has appeared in at least 5 years. This leaves us with roughly 896 counties. We merge this with deposit data and implement regressions that include the county and SEA-year fixed effects. The results are reported in Table 10. Deposits are positively associated with birth rates. A one-standard-deviation change in per capita deposits is associated with 0.18 additional births per 1,000 women in the county, equivalent to roughly 0.5 percent rise in the mean

of the outcome, a quite small change. Although we find evidence of procyclical fertility, the results fail to provide evidence of compositional changes in births. Specifically, there is no change in the share of births to white and black mothers (columns 2 and 3). The marginal effects are statistically insignificant and economically small, as implied by percent changes from the mean of the outcome. For instance, a one-standard-deviation rise in deposits is associated with a three basis-point decrease in the share of births to whites, roughly 0.05 percent change from the mean. If we believe that whites have higher longevity for unobservable reasons, then the associated reduction in their fertility could understate the effects in the main results.

Economic conditions could also affect the survival of fetuses into birth. Fetal death selection is disproportionately higher among males to a degree that sex ratio at birth has been used as a proxy for fetal death (Sanders & Stoecker, 2015). We explore the effects of changes in local economic conditions on sex ratio at birth using the 1940 full-count Census. We report and discuss the results in Appendix D. We find insignificant effects that are very small in magnitude. We further examine selective fertility response of parents to changes in deposits based on maternal education. These results, also reported in Appendix D, do not provide a consistent and strong evidence of selective fertility behavior by maternal education.

7. Mechanism

The results so far suggest that early-life economic conditions have moderate and robust effects on later-life longevity. To establish a candidate mediatory link, we explore the effects on later-life education-income profiles. However, in the 1940 Census, the cohorts of our final sample (born in 1926-1936) had not completed their education. In addition to this issue, post-1940 censuses do not provide county identifiers. To overcome this problem, we use the 1960 census in which we have a below-state geographic identifier: Public Use Microdata Area (PUMA).

PUMA is a Census-defined geographic boundary that identifies places based on their population. In urban areas with a higher population (and population density), a county contains several PUMAs. In rural areas with a lower population, several counties are grouped to form one PUMA. We convert our deposit data into PUMA level by aggregating the deposits for severalcounty PUMAs and assigning similar values to different PUMAs within a county that covers several PUMAs. We then merge this with observations in the 1960 census based on PUMA and birth year. To alleviate the migration issue, we restrict the sample to individuals whose state of birth is the same as state-of-residence in 1960. We also restrict the sample to cohorts born between the years 1928-1936.¹³ We implement regressions that include, in addition to individual covariates, PUMA and state-birth-year fixed effects.¹⁴ The results are reported in Table 11 The results suggest a strong statistical association between per capita deposits in the birth year and educational outcomes and measures of the socioeconomic index. For instance, a one-standard-deviation rise in deposits is associated with 0.04 additional years of schooling (column 1), 57 basis points increase in the probability of any college education (column 2), \$27 higher wage income (column 6), and 0.18 units increase in the socioeconomic score (column 8). We can scale up these effects using changes in state-level income per capita from peak to trough of the Great Depression (years 1929-1933) and its link to deposits as discussed in section 5.5. Such changes in deposits are associated with about 0.13 years drop in years of schooling, roughly 2 percentage-points fall in the probability of college education, and 0.6 units drop in socioeconomic score.

¹³ We use the variable indicating birth year in the 1960 censuses to identify cohorts born between 1928-1936. The reason behind this cohort selection is to have a sample of cohorts similar to the DMF-census-linked sample used in the main analysis of the paper.

¹⁴ Since in many cases PUMA contains several SEAs, we most of the identifying variation comes from PUMA-year level, we avoid using SEA-year fixed effects. Instead, we include state-year fixed effects.

We can use the values reported by similar studies to understand how much of the effects could operate through these channels. Fletcher & Noghanibehambari (2021) explore the effects of new college opening during adolescence on later-life longevity. They find that having a college education raises the age at death by about 1.6 years. Combining this figure with the marginal effect of column 2, one can deduce that a one-standard-deviation increase in deposits raises the age at death by 0.1 months if it solely operates through increases in college education. This number can explain only 6.4 percent of the observed reduced-form effect.¹⁵ In another study to explore the effects of education on mortality, Halpern-Manners et al. (2020) implement a twin-strategy and find that an additional year of schooling is associated with 0.34 years higher age at death. Using the coefficient of column 1, we can infer that, had only the effects operated only through improvements in schooling, a one-standard-deviation rise in deposits leads to 0.15 months increase in longevity, equivalent to roughly 8.8 percent of the reduced-form marginal effect in Table 4.¹⁶

8. Conclusion

The Great Depression was an extraordinary event in the economic history of the United States. From 1929 to 1933, real output contracted by more than 25 percent and the unemployment rate increased from 3.2 percent to 25 percent, reaching the highest levels ever documented. Despite its magnitude, previous literature has found little evidence that the Great Depression affected adult mortality. In this paper we provide new evidence on this link by using local banking deposits, as a proxy for economic conditions and credit market, during in-utero and year of birth can influence

¹⁵ The treatment-on-treated calculation of Fletcher & Noghanibehambari (2021) suggests 1.6 years of increased longevity as a result of college education. We combine this number with the estimated effect of column 2 of Table 11 assuming that the effects solely operate through college education channel. Hence, a one-standard deviation rise in deposits is associated with 0.0057×1.6 years or 0.11 months of additional life. This number is 6.4 percent of the marginal effect of the reduced-form effect of deposits in longevity reported in column 4 of Table 4 (100×0.11/1.7).

¹⁶ This is calculated using 0.34 years effect of Halpern-Manners et al. (2020), column 4 of Table 4 (1.7), and column 1 of Table 11 (0.037), as follows: $\frac{0.34 \times 12 \times 0.037}{1.7} \times 100$

old-age longevity. We find that a one-standard-deviation rise in per capita bank deposits is associated with about 1.7 months higher age at death during old ages. The effect is statistically significant, economically meaningful, and robust across a wide array of specification checks.

A battery of balancing tests rules out significant changes in demographic and family socioeconomic characteristics associated with changes in deposits. Moreover, we fail to find any associations between deposit changes in postnatal ages and later-life longevity suggesting that only conditions in-utero and first year of life is important for later-life longevity. We also argue that endogenous fertility response of parents from different demographic groups does not affect the main results. Additional analysis suggests quite strong associations between bank deposits and retail sale and income per capita, which implies that banking deposits are indeed a reasonable proxy to capture local economic conditions. In addition, we show that improvements in education-income profile during adulthood are potential mechanism channels. However, we argue that between 6-9 percent of the link between early-life deposits and later-life longevity can be explained by modest changes in educational outcomes. These small effects on potential mediatory outcomes suggest that the economic conditions operate through other non-labor-market channels to impact longevity such as changes in health capital that can be detected in old ages.

References

- Acemoglu, D., & Johnson, S. (2007). Disease and development: The effect of life expectancy on economic growth. *Journal of Political Economy*, 115(6), 925–985. https://doi.org/10.1086/529000/SUPPL_FILE/32236DATA.ZIP
- Aizer, A., Eli, S., Ferrie, J., & Muney, A. L. (2016). The Long-Run Impact of Cash Transfers to Poor Families. *American Economic Review*, 106(4), 935–971. https://doi.org/10.1257/AER.20140529
- Aizer, A., Eli, S., Lleras-Muney, A., & Lee, K. (2020). Do Youth Employment Programs Work? Evidence from the New Deal. https://doi.org/10.3386/W27103
- Aizer, A., Stroud, L., & Buka, S. (2016). Maternal stress and child outcomes: Evidence from siblings. *Journal of Human Resources*, 51(3), 523–555. https://doi.org/10.3368/jhr.51.3.0914-6664R
- Almond, D., & Currie, J. (2011). Killing Me Softly: The Fetal Origins Hypothesis. Journal of Economic Perspectives, 25(3), 153–172. https://doi.org/10.1257/JEP.25.3.153
- Almond, D., Currie, J., & Duque, V. (2018). Childhood circumstances and adult outcomes: ActII. *Journal of Economic Literature*, 56(4), 1360–1446.
- Almond, D., Mazumder, B., & van Ewijk, R. (2015). In Utero Ramadan Exposure and Children's Academic Performance. The Economic Journal, 125(589), 1501–1533. https://doi.org/10.1111/ecoj.12168
- Anderson, D. M., Charles, K. K., & Rees, D. I. (2022). Reexamining the Contribution of Public Health Efforts to the Decline in Urban Mortality. *American Economic Journal: Applied Economics*, 14(2), 126–157. https://doi.org/10.1257/APP.20190034

Arthi, V. (2018). "The dust was long in settling": Human capital and the lasting impact of the

American Dust Bowl. *Journal of Economic History*, 78(1), 196–230. https://doi.org/10.1017/S0022050718000074

- Atherwood, S. (2022). Does a prolonged hardship reduce life span? Examining the longevity of young men who lived through the 1930s Great Plains drought. *Population and Environment* 2022, 1–23. https://doi.org/10.1007/S11111-022-00398-W
- Bailey, M., Clay, K., Fishback, P., Haines, M., Kantor, S., Severnini, E., & Wentz, A. (2016).
 U.S. County-Level Natality and Mortality Data, 1915-2007. *Inter-University Consortium for Political and Social Research*. https://doi.org/https://doi.org/10.3886/E100229V4
- Baird, S., Friedman, J., & Schady, N. (2011). Aggregate income shocks and infant mortality in the developing world. *Review of Economics and Statistics*, 93(3), 847–856. https://doi.org/10.1162/REST a 00084
- Barker, D. J. P. (1990). The fetal and infant origins of adult disease. *BMJ: British Medical Journal*, 301(6761), 1111.
- Bengtsson, T., & Broström, G. (2009). Do conditions in early life affect old-age mortality directly and indirectly? Evidence from 19th-century rural Sweden. *Social Science & Medicine*, 68(9), 1583–1590. https://doi.org/10.1016/J.SOCSCIMED.2009.02.020
- Bernanke, B. S. (1983). Non-monetary effects of the financial crisis in the propagation of the Great Depression.
- Black, D. A., Kolesnikova, N., Sanders, S. G., & Taylor, L. J. (2013). Are Children "Normal"? *The Review of Economics and Statistics*, 95(1), 21–33. https://doi.org/10.1162/REST A 00257
- Black, S. E., Devereux, P. J., & Salvanes, K. G. (2007). From the Cradle to the Labor Market? The Effect of Birth Weight on Adult Outcomes. *The Quarterly Journal of Economics*,

122(1), 409-439. https://doi.org/10.1162/qjec.122.1.409

- Bleakley, H. (2007). Disease and Development: Evidence from Hookworm Eradication in the American South. *The Quarterly Journal of Economics*, 122(1), 73–117. https://doi.org/10.1162/QJEC.121.1.73
- Bogue, D. J. (1951). *State economic areas: A description of the procedure used in making a functional grouping of the counties of the United States*. US Government Printing Office.
- Bruckner, T., & Catalano, R. (2007). The sex ratio and age-specific male mortality: evidence for culling in utero. American Journal of Human Biology: The Official Journal of the Human Biology Association, 19(6), 763–773.
- Caruso, G., & Miller, S. (2015). Long run effects and intergenerational transmission of natural disasters: A case study on the 1970 Ancash Earthquake. *Journal of Development Economics*, 117, 134–150. https://doi.org/10.1016/j.jdeveco.2015.07.012
- Catalano, R., Bruckner, T., Anderson, E., & Gould, J. B. (2005). Fetal death sex ratios: a test of the economic stress hypothesis. *International Journal of Epidemiology*, 34(4), 944–948. https://doi.org/10.1093/IJE/DYI081
- Chang, G., Favara, M., & Novella, R. (2022). The origins of cognitive skills and non-cognitive skills: The long-term effect of in-utero rainfall shocks in India. *Economics & Human Biology*, 44, 101089. https://doi.org/10.1016/J.EHB.2021.101089
- Chetty, R., Stepner, M., Abraham, S., Lin, S., Scuderi, B., Turner, N., Bergeron, A., & Cutler, D. (2016). The Association Between Income and Life Expectancy in the United States, 2001-2014. *JAMA*, *315*(16), 1750–1766. https://doi.org/10.1001/JAMA.2016.4226
- Chyn, E. (2018). Moved to Opportunity: The Long-Run Effects of Public Housing Demolition on Children. *American Economic Review*, *108*(10), 3028–3056.

https://doi.org/10.1257/AER.20161352

- Clark, A. E., D'Ambrosio, C., & Rohde, N. (2021). Prenatal economic shocks and birth outcomes in UK cohort data. *Economics & Human Biology*, 41, 100964. https://doi.org/10.1016/J.EHB.2020.100964
- Clay, K., Portnykh, M., & Severnini, E. (2019). The legacy lead deposition in soils and its impact on cognitive function in preschool-aged children in the United States. *Economics & Human Biology*, 33, 181–192. https://doi.org/10.1016/J.EHB.2019.03.001
- Cohen, A., Dehejia, R., & Romanov, D. (2013). Financial Incentives and Fertility. *The Review of Economics and Statistics*, 95(1), 1–20. https://doi.org/10.1162/REST_A_00342
- Costa, D. L. (2015). Health and the economy in the united states from 1750 to the present. In *Journal of Economic Literature* (Vol. 53, Issue 3, pp. 503–570). American Economic Association. https://doi.org/10.1257/jel.53.3.503
- Currie, J. (2009). Healthy, wealthy, and wise: Socioeconomic status, poor health in childhood, and human capital development. *Journal of Economic Literature*, *47*(1), 87–122. https://doi.org/10.1257/jel.47.1.87
- Currie, J., & Rossin-Slater, M. (2015). Early-life origins of life-cycle well-being: research and policy implications. *Journal of Policy Analysis and Management : [The Journal of the Association for Public Policy Analysis and Management]*, 34(1), 208–242. https://doi.org/10.1002/PAM.21805
- Currie, J., & Schwandt, H. (2014). Short- And long-term effects of unemployment on fertility. *Proceedings of the National Academy of Sciences of the United States of America*, 111(41), 14734–14739.

https://doi.org/10.1073/PNAS.1408975111/SUPPL_FILE/PNAS.1408975111.SAPP.PDF

- Cutler, D., Deaton, A., & Lleras-Muney, A. (2006). The Determinants of Mortality. *Journal of Economic Perspectives*, 20(3), 97–120. https://doi.org/10.1257/JEP.20.3.97
- Cutler, D. M., Miller, G., & Norton, D. M. (2007). Evidence on early-life income and late-life health from America's Dust Bowl era. *Proceedings of the National Academy of Sciences*, 104(33), 13244–13249.
- Cutler, D., & Miller, G. (2005). The role of public health improvements in health advances: The twentieth-century United States. *Demography 2005 42:1*, 42(1), 1–22. https://doi.org/10.1353/DEM.2005.0002
- Deaton, A. (2003). *The great escape.Health, wealth, and the oringins os inequality*. Prinncenton University Press.
- Docquier, F. (2004). Income Distribution, Non-convexities and the Fertility–Income Relationship. *Economica*, 71(282), 261–273. https://doi.org/10.1111/J.0013-0427.2004.00369.X
- Duque, V., & Schmitz, L. (2021). The Influence of Early-Life Economic Shocks on Aging Outcomes: Evidence from the U.S. Great Depression. SSRN Electronic Journal. https://doi.org/10.2139/SSRN.3985570
- Fishback, P. V., Haines, M. R., & Kantor, S. (2007). Births, Deaths, and New Deal Relief during the Great Depression. *The Review of Economics and Statistics*, 89(1), 1–14. https://doi.org/10.1162/REST.89.1.1
- Fisher, I. (1933). The Debt-Deflation Theory of Great Depressions. *Econometrica*, 1(4), 337. https://doi.org/10.2307/1907327
- Fletcher, J. M. (2018). New evidence on the impacts of early exposure to the 1918 influenza pandemic on old-age mortality. *Biodemography and Social Biology*, *64*(2), 123–126.

https://doi.org/10.1080/19485565.2018.1501267

- Fletcher, J. M., Green, J. C., & Neidell, M. J. (2010). Long term effects of childhood asthma on adult health. *Journal of Health Economics*, 29(3), 377–387. https://doi.org/10.1016/J.JHEALECO.2010.03.007
- Fletcher, J. M., & Noghanibehambari, H. (2022). The Siren Song of Cicadas: Early-Life Pesticide Exposure and Later-Life Mortality. *Working Paper*.
- Fletcher, J., & Noghanibehambari, H. (2021). *The Effects of Education on Mortality: Evidence Using College Expansions*. https://doi.org/10.3386/W29423
- Fogel, R. W. (1994). The Relevance of Malthus for the Study of Mortality Today: Long-Run Influences on Health, Mortality, Labor Force Participation, and Population Growth. https://doi.org/10.3386/H0054
- Friedman, M., & Schwartz, A. J. (1963). A monetary history of the United States, 1867-1960 (Vol. 16). Princeton University Press.
- Gagnon, A., & Bohnert, N. (2012). Early life socioeconomic conditions in rural areas and oldage mortality in twentieth-century Quebec. *Social Science & Medicine*, 75(8), 1497–1504. https://doi.org/10.1016/J.SOCSCIMED.2012.06.007
- Goldstein, J. R., Alexander, M., Breen, C., Miranda González, A., Menares, F., Osborne, M.,
 Snyder, M., & Yildirim, U. (2021). Censoc Project. In *CenSoc Mortality File: Version 2.0. Berkeley: University of California*. https://censoc.berkeley.edu/data/
- Goodman-Bacon, A. (2021). The Long-Run Effects of Childhood Insurance Coverage: Medicaid Implementation, Adult Health, and Labor Market Outcomes. *American Economic Review*, *111*(8), 2550–2593. https://doi.org/10.1257/AER.20171671

Granados, J. A. T., Roux, A. V. D., Tapia Granados, J. A., & Diez Roux, A. V. (2009). Life and

death during the Great Depression. *Proceedings of the National Academy of Sciences of the United States of America*, 106(41), 17290–17295. https://doi.org/10.1073/pnas.0904491106

- Grimard, F., Laszlo, S., & Lim, W. (2010). Health, aging and childhood socio-economic conditions in Mexico. *Journal of Health Economics*, 29(5), 630–640. https://doi.org/10.1016/J.JHEALECO.2010.07.001
- Halpern-Manners, A., Helgertz, J., Warren, J. R., & Roberts, E. (2020). The Effects of Education on Mortality: Evidence From Linked U.S. Census and Administrative Mortality Data. *Demography*, 57(4), 1513–1541. https://doi.org/10.1007/S13524-020-00892-6
- Hayward, M. D., & Gorman, B. K. (2004). The long arm of childhood: The influence of earlylife social conditions on men's mortality. *Demography 2004 41:1*, 41(1), 87–107. https://doi.org/10.1353/DEM.2004.0005
- Hornbeck, R. (2011). Replication data for: The Enduring Impact of the American Dust Bowl: Short and Long-run Adjustments to Environmental Catastrophe. Harvard Dataverse. https://doi.org/10.7910/DVN/CLRPAV
- Hoynes, H., Schanzenbach, D. W., & Almond, D. (2016). Long-run impacts of childhood access to the safety net. *American Economic Review*, 106(4), 903–934. https://doi.org/10.1257/aer.20130375
- James, W. H., & Grech, V. (2017). A review of the established and suspected causes of variations in human sex ratio at birth. *Early Human Development*, 109, 50–56. https://doi.org/10.1016/J.EARLHUMDEV.2017.03.002
- Lindeboom, M., Portrait, F., & Van Den Berg, G. J. (2010). Long-run effects on longevity of a nutritional shock early in life: The Dutch Potato famine of 1846–1847. *Journal of Health Economics*, 29(5), 617–629. https://doi.org/10.1016/J.JHEALECO.2010.06.001

- Ludwig, J., & Miller, D. L. (2007). Does Head Start Improve Children's Life Chances? Evidence from a Regression Discontinuity Design. *The Quarterly Journal of Economics*, 122(1), 159– 208. https://doi.org/10.1162/QJEC.122.1.159
- Majid, M. F. (2015). The persistent effects of in utero nutrition shocks over the life cycle:
 Evidence from Ramadan fasting. *Journal of Development Economics*, *117*, 48–57.
 https://doi.org/10.1016/j.jdeveco.2015.06.006
- Manson, S., Schroeder, J., Riper, D. Van, Ruggles, S., Others, Van Riper, D., Ruggles, S., &
 Others. (2017). IPUMS National Historical Geographic Information System: Version 12.0
 [Database]. *Minneapolis: University of Minnesota*, 39.
- Miller, S., & Wherry, L. R. (2019). The Long-Term Effects of Early Life Medicaid Coverage. Journal of Human Resources, 54(3), 785–824. https://doi.org/10.3368/jhr.54.3.0816.8173r1
- Montez, J. K., & Hayward, M. D. (2011). Early Life Conditions and Later Life Mortality. *International Handbook of Adult Mortality*, 187–206. https://doi.org/10.1007/978-90-481-9996-9_9
- Morse, A., & Luke, N. (2021). Foetal loss and feminine sex ratios at birth in sub-Saharan Africa. *Population Studies*, 75(2), 239–254. https://doi.org/10.1080/00324728.2021.1877793
- Myrskylä, M. (2010a). The effects of shocks in early life mortality on later life expectancy and mortality compression: A cohort analysis. *Demographic Research*, 22, 289–320. https://doi.org/10.4054/DemRes.2010.22.12
- Myrskylä, M. (2010b). The Relative Effects of Shocks in Early- and Later-Life Conditions on Mortality. *Population and Development Review*, *36*(4), 803–829. https://doi.org/10.1111/j.1728-4457.2010.00358.x

Noghanibehambari, H., & Engelman, M. (2022). Social insurance programs and later-life

mortality: Evidence from new deal relief spending. *Journal of Health Economics*, *86*, 102690. https://doi.org/10.1016/J.JHEALECO.2022.102690

Noghanibehambari, H., & Fletcher, J. (2022). *Dust to Feed, Dust to Grey: The Effect of In-Utero Exposure to Dust Bowl on Old-Age Longevity.*

Persson, P., & Rossin-Slater, M. (2018). Family ruptures, stress, and the mental health of the next generation. *American Economic Review*, 108(4–5), 1214–1252. https://doi.org/10.1257/aer.20141406

- Preston, S. H. (1975). The Changing Relation between Mortality and level of Economic Development. *Population Studies*, 29(2), 231–248. https://doi.org/10.1080/00324728.1975.10410201
- Quintana-Domeque, C., & Ródenas-Serrano, P. (2017). The hidden costs of terrorism: The effects on health at birth. *Journal of Health Economics*, 56, 47–60. https://doi.org/10.1016/j.jhealeco.2017.08.006
- Rosa, M. J., Nentin, F., Bosquet Enlow, M., Hacker, M. R., Pollas, N., Coull, B., & Wright, R. J. (2019). Sex-specific associations between prenatal negative life events and birth outcomes. *Stress*, *22*(6), 647–653.
 https://doi.org/10.1080/10253890.2019.1608944/SUPPL_FILE/ISTS_A_1608944_SM4672
 .DOCX
- Ruggles, S., Flood, S., Goeken, R., Grover, J., & Meyer, E. (2020). IPUMS USA: Version 10.0 [dataset]. *Minneapolis, MN: IPUMS*. https://doi.org/10.18128/D010.V10.0
- Sanders, N. J., & Stoecker, C. (2015). Where have all the young men gone? Using sex ratios to measure fetal death rates. *Journal of Health Economics*, 41, 30–45. https://doi.org/10.1016/J.JHEALECO.2014.12.005

- Schaller, J. (2016). Booms, busts, and fertility: Testing the becker model using gender-specific labor demand. *Journal of Human Resources*, *51*(1), 1–29. https://doi.org/10.3368/jhr.51.1.1
- Schaller, J., Fishback, P., & Marquardt, K. (2020). Local Economic Conditions and Fertility from the Great Depression through the Great Recession. *AEA Papers and Proceedings*, *110*, 236–240. https://doi.org/10.1257/PANDP.20201109
- Schmitz, L., & Duque, V. (2022). In Utero Exposure to the Great Depression is Reflected in Late-Life Epigenetic Aging Signatures. https://doi.org/10.1101/2022.05.18.22275258
- Schoeps, A., van Ewijk, R., Kynast-Wolf, G., Nebié, E., Zabré, P., Sié, A., & Gabrysch, S. (2018). Ramadan Exposure In Utero and Child Mortality in Burkina Faso: Analysis of a Population-Based Cohort Including 41,025 Children. *American Journal of Epidemiology*, *187*(10), 2085–2092. https://doi.org/10.1093/AJE/KWY091
- Scholte, R. S., Van Den Berg, G. J., & Lindeboom, M. (2015). Long-run effects of gestation during the Dutch Hunger Winter famine on labor market and hospitalization outcomes. *Journal of Health Economics*, 39, 17–30.

https://doi.org/10.1016/J.JHEALECO.2014.10.002

- Shah, M., & Steinberg, B. M. (2017). Drought of opportunities: Contemporaneous and long-term impacts of rainfall shocks on human capital. *Journal of Political Economy*, 125(2), 527– 561. https://doi.org/10.1086/690828/SUPPL_FILE/2013254DATA.ZIP
- Smith, A. H., Marshall, G., Yuan, Y., Liaw, J., Ferreccio, C., & Steinmaus, C. (2011). Evidence From Chile That Arsenic in Drinking Water May Increase Mortality From Pulmonary Tuberculosis. *American Journal of Epidemiology*, *173*(4), 414–420. https://doi.org/10.1093/AJE/KWQ383

Steptoe, A., & Zaninotto, P. (2020). Lower socioeconomic status and the acceleration of aging:
An outcome-wide analysis. *Proceedings of the National Academy of Sciences of the United States of America*, *117*(26), 14911–14917. https://doi.org/10.1073/PNAS.1915741117

- Stuckler, D., Meissner, C., Fishback, P., Basu, S., & McKee, M. (2012). Banking crises and mortality during the Great Depression: evidence from US urban populations, 1929–1937. J Epidemiol Community Health, 66(5), 410–419. https://doi.org/10.1136/JECH.2010.121376
- Van Den Berg, G. J., Doblhammer, G., & Christensen, K. (2009). Exogenous determinants of early-life conditions, and mortality later in life. *Social Science & Medicine*, 68(9), 1591– 1598. https://doi.org/10.1016/J.SOCSCIMED.2009.02.007
- Van Den Berg, G. J., Gupta, S., van den Berg, G. J., & Gupta, S. (2015). The role of marriage in the causal pathway from economic conditions early in life to mortality. *Journal of Health Economics*, 40, 141–158. https://doi.org/10.1016/j.jhealeco.2014.02.004
- Van Den Berg, G. J., Lindeboom, M., Portrait, F., Berg, G. J. Van Den, Lindeboom, M., Portrait, F., den Berg, G. J., Lindeboom, M., & Portrait, F. (2006). Economic Conditions Early in Life and Individual Mortality. *American Economic Review*, 96(1), 290–302. https://doi.org/10.1257/000282806776157740
- Wang, J., Gao, Z. Y., Yan, J., Ying, X. L., Tong, S. L., & Yan, C. H. (2017). Sex differences in the effects of prenatal lead exposure on birth outcomes. *Environmental Pollution*, 225, 193– 200. https://doi.org/10.1016/J.ENVPOL.2017.03.031
- Weinberg, J., Sliwowska, J. H., Lan, N., & Hellemans, K. G. C. (2008). Prenatal Alcohol Exposure: Foetal Programming, the Hypothalamic-Pituitary-Adrenal Axis and Sex Differences in Outcome. *Journal of Neuroendocrinology*, *20*(4), 470–488. https://doi.org/10.1111/J.1365-2826.2008.01669.X

Yamashita, N., & Trinh, T. A. (2022). Effects of prenatal exposure to abnormal rainfall on

cognitive development in Vietnam. *Population and Environment*, 43(3), 346–366. https://doi.org/10.1007/S11111-021-00394-6/TABLES/6

Tables

Table 1 - Summary Statistics

Table 1 - Summary Statistics									
Variable	Observations	Mean	Std. Dev.	Min	Max				
Death Age (Months)	1221113	772.66643	104.36921	457	959				
Death Age (Years)	1221113	63.88002	8.71601	38	79				
Birth Year	1221113	1930.1929	3.10607	1926	1936				
Death Year	1221113	1994.5757	8.31481	1975	2005				
Deposits per Capita (STD)*	1221113	0	1	58422	50.79202				
Deposits per Capita	1221113	330.91799	566.43036	0	29101.057				
Total Deposits (\$1B)**	1221113	.30424	1.08423	0	12.59437				
Drop in Deposits>5% & $\leq 10\%^{***}$	1221113	.10077	.30102	0	1				
Drop in Deposits>10%	1221113	.23087	.42139	0	1				
White	1221113	.91466	.27939	0	1				
Black	1221113	.08022	.27164	0	1				
Father SEI 1 st Quartile	1221113	.23911	.42654	0	1				
Father SEI 2 nd Quartile	1221113	.22718	.41901	0	1				
Father SEI 3 rd Quartile	1221113	.22679	.41876	0	1				
Father SEI 4 th Quartile	1221113	.20241	.40179	0	1				
Father SEI Missing	1221113	.04182	.20017	0	1				
Mother Education <hs< td=""><td>1221113</td><td>.60398</td><td>.48907</td><td>0</td><td>1</td></hs<>	1221113	.60398	.48907	0	1				
Mother Education =HS	1221113	.27552	.44678	0	1				
Mother Education >HS	1221113	.04961	.21714	0	1				
Mother Education Missing	1221113	.07089	.25664	0	1				
Work-Related per Capita Relief	1221113	5.00054	9.49932	0	376.46896				
Spending									
Average Homeownership	1221113	.50192	.13717	.02538	.90763				
Share of Literate	1221113	.81183	.1943	0	1				
Average Occupational Income	1221113	23.73044	4.10521	11.78472	29.7175				
Score									
Share of Married Females	1221113	.60946	.03199	.28999	.73985				
County-Level Data:									
Retail Sale per Capita	20258	174.9985	98.5671	0	782.43945				
State-Level Data:									
Income per Capita	376	445.60084	195.13996	122.98853	1151.4171				
The 1960-Census Data:									
Years of Schooling	403493	7.89761	2.93819	0	15				
Educ>1 Year of College	403493	.17413	.37923	0	1				
Educ>2 Year of College	403493	.13247	.339	0	1				
Educ>3 Year of College	403493	.0963	.295	0	1				
Educ>4 Year of College	403493	.07397	.26172	0	1				
Wage and Salary Income	403493	1622.6333	1873.754	0	19866.486				
Total Income	403493	1879.7341	2068.2311	-7906.8618	19866.486				
Socioeconomic Index	342766	34.44022	21.7157	3	96				
Occupational Prestigious Score	340401	35.80639	12.37729	9.3	81.5				
Occupational Education Score	340346	18.55855	22.3739	.8	100				

Notes. The statistics are weighted using county-level mean population. * The abbreviation STD represents standardized variable. ** Total deposits measure average county-year total deposits in all banks reported in the data. Its unit in this table is billions of dollars.

*** Percent drop is the drop in a county's deposits with respect to the county's previous year's deposits.

	Outcomes:						
	White	Mother's Schooling	Mother's Education Missing	Father's Schooling	Father's Education Missing	Father's SEI	Father's SEI Missing
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Deposits per Capita	00254	.012	.00072	.00918	.002***	10234	.00237**
(STD)	(.00161)	(.01309)	(.00132)	(.01603)	(.00068)	(.07191)	(.00098)
Observations	1221113	1133633	1221113	1068262	1221113	1040576	1221113
R-squared	.11896	.08817	.04105	.07459	.04562	.0515	.02668
Mean DV	0.944	8.125	0.063	8.083	0.016	32.052	0.043
%Change	-0.270	0.148	1.138	0.114	12.513	-0.319	5.504
County FE	✓	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	✓
SEA-by-Birth-Year FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark

Table 2 - Balancing Tests

Notes. Standard errors, clustered at the county-level, are in parentheses. The regressions are weighted using county-level mean of population over the sample period. The "%Change" values are calculated using the estimated marginal effects of row 1 divided by the mean of dependent variables reported in row 4.

*** p<0.01, ** p<0.05, * p<0.1

	Outcome: Successful DMF-1940-Census Merging; Subsamples:					
-			Mother Education	Father's SEI below		
	Full Sample	Whites	less than High	Median		
_			School			
	(1)	(2)	(3)	(4)		
Deposits per	.00008	.00007	00006	.00011		
Capita (STD)	(.00009)	(.00009)	(.00013)	(.00009)		
Observations	22285131	19603408	12940827	10365020		
R-squared	.06454	.06571	.06851	.06721		
Mean DV	0.057	0.058	0.060	0.058		
%Change	0.148	0.125	-0.107	0.188		
County FE	\checkmark	\checkmark	\checkmark	\checkmark		
SEA-by-Birth-	\checkmark	\checkmark	\checkmark	\checkmark		
Year FE						
Family Controls	\checkmark	\checkmark	\checkmark	\checkmark		
County Controls	\checkmark	\checkmark	\checkmark	\checkmark		

Table 3 - The Correlation between Successful Merging of DMF-Census data and Per Capita Deposits

Notes. Standard errors, clustered at the county-level, are in parentheses. The regressions are weighted using county-level mean of population over the sample period. The "%Change" values are calculated using the estimated marginal effects of row 1 divided by the mean of dependent variables reported in row 4.

*** p<0.01, ** p<0.05, * p<0.1

	Outcome: Death Age (Months)				
	(1)	(2)	(3)	(4)	
Demonita non Consite (STD)	1.55965***	1.58676***	1.59411***	1.72552**	
Deposits per Capita (STD)	(.57764)	(.59254)	(.585)	(.72026)	
Observations	1221113	1221113	1221113	1221113	
R-squared	.10219	.10312	.1036	.1036	
Mean DV	772.939	772.939	772.939	772.939	
%Change	0.202	0.205	0.206	0.223	
County FE	\checkmark	\checkmark	\checkmark	\checkmark	
SEA-by-Birth-Year FE	\checkmark	\checkmark	\checkmark	\checkmark	
Individual Controls		\checkmark	\checkmark	\checkmark	
Family Controls			\checkmark	\checkmark	
County Controls				\checkmark	

Table 4 - Main Results: The Association between in-Utero Bank Deposits and Old-Age Longevity

Notes. Standard errors, clustered at the county-level, are in parentheses. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period. The "%Change" values are calculated using the estimated marginal effects of row 1 divided by the mean of dependent variables reported in row 4. *** p < 0.01, ** p < 0.05, * p < 0.1

	Outcome: Death Age (Months)				
	(1)	(2)	(3)		
Two Years Pre-Prenatal	54054	54144	43639		
Two Years Pre-Prenatal	(.65818)	(.66487)	(.82259)		
During In Litera	2.14054**	2.13353**	1.96145**		
During In Utero	(1.07014)	(1.07366)	(.95119)		
True Veens Destructed	82156	83721	65997		
Two Years Postnatal	(.67746)	(.68141)	(.70487)		
Observations	1142972	1142972	1142972		
R-squared	.09109	.09161	.09161		
Mean DV	776.140	776.140	776.140		
County FE	\checkmark	\checkmark	\checkmark		
SEA-by-Birth-Year FE	\checkmark	\checkmark	\checkmark		
Individual Controls	\checkmark	\checkmark	\checkmark		
Family Controls		\checkmark	\checkmark		
County Controls			\checkmark		

Table 5 - Assigning Deposits per Capita at Pre-prenatal, Prenatal, and Postnatal Periods

Notes. Standard errors, clustered at the county-level, are in parentheses. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period. The "%Change" values are calculated using the estimated marginal effects of row 1 divided by the mean of dependent variables reported in row 4.

*** p<0.01, ** p<0.05, * p<0.1

	Outcome: Death Age (Months)							
	(1)	(2)	(3)	(4)	(5)	(6)	(7	
State-Level	9.3903***	1.3029*						
Income Per	(1.26957)	(.74357)						
Capita	· · · ·							
State-Level			32.51253***	6.20618				
Deposits per			(6.58206)	(6.03937)				
Capita				. ,				
County-Level					.41644	53373***	2.4080	
Deposits per					(.83148)	(.04354)	(.893	
Capita							×	
Observations	771158	771158	771158	771158	771158	771158	7711	
R-Squared	.01005	.05348	.0036	.05348	.00002	.05241	.052	
Mean DV	754.850	754.850	754.850	754.850	755.112	755.112	755.	
%Change	1.244	0.173	4.307	0.822	0.055	-0.071	0.3	
State FE		\checkmark		\checkmark		\checkmark		
County FE							\checkmark	
Birth Year FE		\checkmark		\checkmark		\checkmark	\checkmark	
SEA-by-Birth-								
Year FE								
Region-by-Birth-		\checkmark		\checkmark		\checkmark		
Year FE		•		•		•		

Table 6 - Comparing the Effects of State-Level Economic Conditions with the Effects of Local Economic Conditions on Longevity

Notes. Standard errors are in parentheses. Standard errors of columns 1-4 are clustered at the state-level. Standard errors of columns 5-8 are clustered at the county-level. The regres mean of population over the sample period. *** p<0.01, ** p<0.05, * p<0.1

	County-Level Outcome: Per Capita Retail Sale (STD)				State-Level Outcome				
-				Per Capita Income (STD)					
_	(1)	(2)	(3)	(4)	(5)	(6)			
D :	.86484***	.18034***	.14568***	.72179***	.38056***	.41758***			
Deposits per Capita (STD)	(.2771)	(.06852)	(.05264)	(.17681)	(.07804)	(.07123)			
Observations	20237	20236	19574	376	376	376			
R-squared	.31861	.97957	.99003	.50444	.96404	.98016			
Mean DV	0.000	0.000	-0.009	0.000	0.000	0.000			
County FE		\checkmark	\checkmark						
State FE					\checkmark	\checkmark			
Year FE		\checkmark	\checkmark		\checkmark	\checkmark			
SEA-by-Year			\checkmark						
FE									
Region-by- Year FE						\checkmark			

 Table 7 - The Correlations between Deposits per Capita and other Economic Variables (1929-1936)

Notes. Standard errors, reported in parentheses, are clustered at the county-level for columns 1-2 and state-level for columns 3-4. The regressions are weighted using countylevel/state-level mean of population. *** p<0.01, ** p<0.05, * p<0.1

	Column 4 Table 4	Adding County-by- Individual FE	Adding County-by- Parental Covariates FE
	(1)	(2)	(3)
	1.72552**	1.72981**	1.79781**
Deposits per Capita (STD)	(.72026)	(.71853)	(.70769)
Observations	1221113	1220764	1220880
R-squared	.1036	.10389	.10501
	Adding County-by-Birth- Year Trend	Adding Birth-Month and Death-Month FE	Outcome: Log Age at Death
	(4)	(5)	(6)
	2.39254***	1.72548**	.00244**
Deposits per Capita (STD)	(.59011)	(.70252)	(.00103)
Observations	1221113	1221113	1221113
R-squared	.10396	.10447	.09848
	Outcome: Death Age > 70 Years	Outcome: Death Age > 65 Years	Two-Way Clustering SE at County by State-Birth- Year Level
	(7)	(8)	(9)
	.00625***	.00531*	1.70281**
Deposits per Capita (STD)	(.00157)	(.00309)	(.74797)
Observations	1221113	1221113	1218545
R-squared	.13318	.06653	.10269

Table 8 - Robustness Checks

Notes. Standard errors, clustered at the county-level (except for columns 9-10), are in parentheses. All regressions include county fixed effects, birth-year fixed effects, county-division fixed effects, and county trend. All regressions include individual, parental, and county covariates. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period.

*** p<0.01, ** p<0.05, * p<0.1

	Outcome: Death Age (Months)			
	(1)	(3)		
Total Deposits (STD)	2.19549*** (.82583)			
Drop in Deposits>5% and <10%		60259** (.28177)		
Drop in Deposits>10%		59231* (.33466)		
Observations	1206663	1208367		
R-squared	.1035	.10352		
Mean DV	773.034	773.021		

Table 9 - Alternative Measures

Notes. Standard errors, clustered at the county-level, are in parentheses. All regressions include county fixed effects, birth-year fixed effects, county-division fixed effects, and county trend. All regressions include individual, parental, and county covariates. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period. *** p < 0.01, ** p < 0.05, * p < 0.1

	Outcomes:					
	Births per 1000 Women	Share of Births to Whites	Share of Births to Blacks			
	(2)	(3)	(4)			
Deposits per Capita (STD)	.18997**	00036	.00022			
	(.08947)	(.00112)	(.00105)			
Observations	8365	8365	8365			
R-squared	.92043	.97239	.98336			
Mean DV	36.824	0.701	0.297			
%Change	0.516	-0.052	0.075			
County FE	\checkmark	\checkmark	\checkmark			
Year FE	\checkmark	\checkmark	\checkmark			
SEA-Year FE	\checkmark	\checkmark	\checkmark			

Table 10 - The Association between	n Deposits per	Capita and Births Rates
------------------------------------	----------------	-------------------------

Notes. Standard errors, reported in parentheses, are clustered at the county-level. The regressions are weighted using county-level mean of population over the sample period. *** p<0.01, ** p<0.05, * p<0.1

	Outcomes:								
	Years of Schooling	Education≥ 1 Year of College	Education≥ 2 Years of College	Education≥ 3 Years of College	Education≥ 4 Years of College	Wage Income	Total personal Income	Socioecono mic Index	Occ Pre
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Deposits per Capita	.03648**	.00573***	.00477***	.00353**	.004***	27.07686***	30.1783***	.17735*	.0
(STD)	(.01797)	(.00148)	(.00149)	(.00144)	(.00146)	(6.77593)	(8.75529)	(.09199)	(
Observations	310975	310975	310975	310975	310975	310975	310975	269899	2
R-squared	.12647	.04356	.03782	.03092	.02832	.39537	.43729	.12008	
Mean DV	7.916	0.179	0.138	0.102	0.080	1766.282	2032.139	35.233	3
%Change	0.461	3.202	3.457	3.456	5.005	1.533	1.485	0.503	
Puma-County FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	√	
State-Year FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	

 Table 11 - Exploring the Mechanisms of Impact Using 1960 Census

Notes. Standard errors, reported in parentheses, are clustered at the county-Puma-level. The regressions are weighted using county-Puma-level mean of population over the sample period. *** p < 0.01, ** p < 0.05, * p < 0.1

Figures



Figure 1 - Geographic Distribution of Variables

Notes. The colors in the map are based on the county's quartile rank in the nation's distribution of the respective variable.



Figure 2 - Time-Series Evolution of Per Capita Bank Deposits across Census Regions



Figure 3 - Density Distribution of Age at Death by County above/below Median Bank Deposit Per Capita

Appendix A

In Appendix Table A-1, we replicate the main results across subsamples of whites and blacks. We observe relatively larger coefficients among whites. A one-standard-deviation rise in bank deposits per capita during in-utero is associated with 4.9 and 1.9 months higher longevity among blacks and whites, respectively.

	Outcome: Age at Death (Months)		
	Blacks	Whites	
	(1)	(2)	
Deposits per Capita (STD)	4.85125**	1.93562***	
	(2.42706)	(.70076)	
Observations	95690	1116246	
R-squared	.13416	.10225	
Mean DV	754.521	773.914	
%Change	0.643	0.250	
County FE	\checkmark	\checkmark	
SEA-by-Birth-Year FE	\checkmark	\checkmark	
Individual Controls	\checkmark	\checkmark	
Family Controls	\checkmark	\checkmark	
County Controls	\checkmark	\checkmark	

Notes. Standard errors, clustered at the county-level, are in parentheses. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period. *** p < 0.01, ** p < 0.05, * p < 0.1

Appendix B

In this appendix, we replicate the main results using Numident death records (1988-2005) as an alternative data source. We start by replicating the full specification of column 4 of Table 4 in the first column of Appendix Table B-1. We then restrict DMF data to the coverage years of Numident data, i.e., years 1988-2005, in column 2. We reach a significant coefficient of 0.65, about 38 percent of the size of the main results. Using male deaths in Numident, we observe an effect of 0.8 months (column 3). Comparing columns 2 and 3, we observe a relatively larger effect in Numident data. However, focusing on female death records reveal an insignificant effect of 0.3 months (column 4). Pooling both genders in Numident data suggests a significant effect of 0.55 months (column 5).

		Outcome: Age at Death (Months)				
	DMF, Males, Death Years 1975-2005	DMF, Males, Death Years 1988-2005	Numident, Males, Death Years 1988- 2005	Numident, Females, Death Years 1988- 2005	Numident, All, Death Years 1988-2005	
	(1)	(2)	(3)	(4)	(5)	
Deposits per Capita (STD)	1.72552**	.64944***	.79878***	.27035	.54779***	
	(.72026)	(.24692)	(.18345)	(.31548)	(.16828)	
Observations	1221113	938470	1336616	976853	2327996	
R-squared	.1036	.2605	.26177	.27642	.26492	
Mean DV	772.939	819.901	820.745	827.705	823.705	
%Change	0.223	0.079	0.097	0.033	0.067	
County FE	✓	\checkmark	\checkmark	\checkmark	\checkmark	
SEA-by-Birth-Year FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
Individual Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
Family Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
County Controls	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	

Appendix Table B-1 - Sensitivity of Results to Death Window Coverage

Notes. Standard errors, clustered at the county-level, are in parentheses. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period. *** p < 0.01, ** p < 0.05, * p < 0.1

Appendix C

One potential heterogeneity is across counties with differential exposure to the Dust Bowl and the resulting soil erosions. Although Dust Bowl started mainly after 1935, they could impose selection bias to our results if economic conditions have differential impact on health of newborns in exposed and unexposed counties. We use data from Hornbeck (2011) to distinguish counties with high and medium soil erosion due to the Dust Bowl from all other counties. We then replicate the main results across exposed and unexposed counties. The results are reported in Appendix Table C-1. A one-standard-deviation increase in bank deposits per capita during the prenatal development is associated with 9.1 and 1.7 months higher longevity in exposed and unexposed counties, respectively. However, the effect of exposed counties is insignificant. Therefore, we rule out the concern that behavior of Dust Bowl exposed counties drive the main results.

	Outcome: Age a	Outcome: Age at Death (Months)		
	Dust Bowl Counties	All Other Counties		
	(1)	(2)		
Deposits per Capita (STD)	9.05849	1.73333**		
	(10.3922)	(.72988)		
Observations	174684	1043160		
R-squared	.12838	.103		
Mean DV	771.158	772.973		
%Change	1.175	0.224		
No of Counties	776	2278		
County FE	\checkmark	\checkmark		
SEA-by-Birth-Year FE	\checkmark	\checkmark		
Individual Controls	\checkmark	\checkmark		
Family Controls	\checkmark	\checkmark		
County Controls	\checkmark	\checkmark		

Appendix Table C-1 - Heterogeneity of the Results across Counties by their Exposure to Dust Bowl

Notes. Standard errors, clustered at the county-level, are in parentheses. Individual controls include race and gender dummies. Parental controls include father's socioeconomic status dummies and mother's education dummies. County-by-birth-year covariates include share of white-collar workers, share of blue-collar workers, share of farmers, and share of literate people. The regressions are weighted using county-level mean of population over the sample period. *** p<0.01, ** p<0.05, * p<0.1

Appendix D

In section 6, we discussed potential fertility responses to local economic conditions. We found evidence of procyclicality in the fertility behavior of parents but did not find evidence for changes in the composition of births based on race. One source of changes in the composition of births is selection by fetal deaths. Several studies suggest that conditions prior to and during pregnancy affect fetal death and that the incidence of fetal death is disproportionately higher among male fetuses (Quintana-Domeque & Ródenas-Serrano, 2017; Sanders & Stoecker, 2015). Therefore, fetal death and sex ratio at birth are strongly connected (Bruckner & Catalano, 2007; Catalano et al., 2005; James & Grech, 2017; Morse & Luke, 2021). We can also assess the selective survival of fetuses during pregnancy by evaluating the sex ratio at birth. However, the natality data over the period of the study is limited to race (white and black) disaggregation and does not provide more information on the share of females in births at the county level. To overcome this limitation and assess this source of selection, we turn to the 1940 full-count Census. We limit the data to cohorts born between 1926-1936 to have the same cohorts as the final sample of the paper. We use the same procedure as described in section 3 to infer the county of birth. We use the information on gender to define a dummy indicating female. We then collapse the data at the county-year-ofbirth level. The idea is that, conditional on survival up to 1940, the observed sex ratios in the 1940 Census could be used as a proxy for sex ratio at birth. We then merge this data with bank deposit data at the county-year level and implement regressions similar to equation 1. The results are reported in Appendix Table D-1 across columns for different specifications. We observe statistically insignificant effects that are very small in magnitude. In the fully parametrized model of column 3 that includes controls for the county-year share of different races, county fixed effects, and SEA-year fixed effects, a one-standard-deviation rise in deposits is associated with a 0.6 basispoints rise in sex ratio at birth (share of females to males), equivalent to 0.012 percent change with respect to the mean of the outcome.

Another interesting fertility selection is differential births response to economic conditions by maternal education. Since the natality data used in section 6 lacks information on maternal education, we use the full-count 1940 Census, focus on the 1926-1936 cohorts, and infer the county of birth as described in section 3. We then collapse the sample at the county-year-of-birth level and merge it with county-level deposit data. We assume that, conditional on survival into 1940, the full-count Census reveals the universe of births and can be used to examine the selective fertility response to deposits by maternal education. In so doing, we regress measures of maternal education from the collapsed Census data on per capita deposits, conditional on county and SEAyear fixed effects. The results are reported in Appendix Table D-2 for a continuous measure of the mother's schooling (column 1) and three binary measures of the mother's education (columns 2-4). The estimated effects are very small in magnitude. For instance, a one-standard-deviation change in deposits is associated with roughly 0.04 percent change in the mother's schooling (from the mean of the outcome). Except for mother education less than high school, the coefficients are statistically insignificant.

		Outcome: Sex Ratio)
	(1)	(2)	(3)
	.00013	.00014	.00006
Deposits per Capita (STD)	(.00018)	(.00015)	(.00022)
Observations	33468	33468	32400
R-squared	.12232	.12365	.23157
Mean DV	0.493	0.493	0.493
%Change	0.026	0.028	0.012
County and Year FE	\checkmark	\checkmark	\checkmark
Controls		\checkmark	\checkmark
SEA-by-Year FE			\checkmark

Appendix Table D-1 - Deposits and Sex Ratio at Birth Using 1940 Full-Count Census

Notes. Standard errors, clustered at the county-level, are in parentheses. Controls include share of whites and blacks in each cell. *** p<0.01, ** p<0.05, * p<0.1

		Outcomes:			
	Mother's Years	Mother's	Mother's	Mother's	
	of Schooling	Education < HS	Education = HS	Education > HS	
	(1)	(2)	(3)	(4)	
Deposits per Capita (STD)	00293	.00067*	00079	00008	
	(.00251)	(.00037)	(.0005)	(.00008)	
Observations	32373	32400	32400	32400	
R-squared	.98387	.98448	.98181	.93932	
Mean DV	8.189	0.568	0.292	0.059	
%Change	-0.036	0.118	-0.271	-0.137	
County and Year FE	\checkmark	\checkmark	\checkmark	\checkmark	
Controls	\checkmark	\checkmark	\checkmark	\checkmark	
SEA-by-Year FE	\checkmark	\checkmark	\checkmark	\checkmark	

Appendix Table D-2 - Deposits and Mothers' Schooling Using 1940 Full-Count Census

Notes. Standard errors, clustered at the county-level, are in parentheses. Controls include share of whites and blacks in each cell. *** p < 0.01, ** p < 0.05, * p < 0.1